## PEER REVIEW HISTORY

BMJ Open publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form (http://bmjopen.bmj.com/site/about/resources/checklist.pdf) and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

## **ARTICLE DETAILS**

TITLE (PROVISIONAL)	Development of machine learning support for reading whole body diffusion weighted magnetic resonance imaging (WB-MRI) in myeloma for the detection and quantification of the extent of disease before and after treatment (MALIMAR): protocol for a cross-sectional diagnostic test accuracy study
AUTHORS	Satchwell, Laura; Wedlake, Linda; Greenlay, Emily; Li, Xingfeng; Messiou, Christina; Glocker, Ben; Barwick, Tara; Barfoot, Theodore; Doran, Simon; Leach, Martin O; Koh, Dow Mu; Kaiser, Martin; Winzeck, Stefan; Qaiser, Talha; Aboagye, Eric; ROCKALL, ANDREA

This article was not externally reviewed at BMJ Open. Protocols that have been independently assessed prior to submission to BMJ Open are usually fast-tracked to publication on the grounds that further substantial changes will not be possible. This independent assessment will usually be external, independent review for both a substantial grant award from a non-commercial or government funder and ethics approval.